Enhanced neoplasia detection in chronic ulcerative colitis: the ENDCaP-C diagnostic accuracy study

Ashish Awasthi,¹ Jamie Barbour,² Andrew Beggs,³ Pradeep Bhandari,⁴ Daniel Blakeway,³ Matthew Brookes,⁵ James Brown,⁶ Matthew Brown,⁷ Germaine Caldwell,³ Samuel Clokie,⁸ Ben Colleypriest,⁹ Abby Conlin,¹⁰ Shanika de Silva,¹¹ John de Caestecker,¹² Jonathan Deeks,^{6,13} Anjan Dhar,¹⁴ Mark Dilworth,³ Edward Fogden,¹⁵ Stephen Foley,¹⁶ Deb Ghosh,¹⁷ Leonie Grellier,¹⁸ Ailsa Hart,¹⁹ Syed Samiul Hoque,²⁰ Marietta Iacucci,²¹ Tariq Iqbal,^{3,21} Jonathan James,³ Mark Jarvis,²² Anthoor Jayaprakash,²³ Satish Keshav,²⁴ Laura Magill,⁶ Glenn Matthews,³ Joel Mawdsley,²⁵ Simon McLaughlin,²⁶ Samir Mehta,⁶ Kevin Monahan,²⁵ Dion Morton,^{3,21}* Senthil Murugesan,²⁷ Miles Parkes,²⁸ Valerie Pestinger,³ Chris Probert,²⁹ Arvind Ramadas,³⁰ Alessandro Rettino,⁸ Shaji Sebastian,³¹ Naveen Sharma,^{21,32} Michael Griffiths,⁸ Joanne Stockton,³ Venkat Subramanian,³³ Nigel Suggett,²¹ Philippe Taniere,²¹ Julian Teare,³⁴ Ajay M Verma³⁵ and Yvonne Wallis⁸ on behalf of the ENDCaP-C **Collaborative Group**

¹Department of Gastroenterology, Manor Hospital, Walsall, UK ²Department of Inflammatory Bowel Disease, Queen Elizabeth Hospital, Gateshead, UK ³Institute of Cancer and Genomic Science, University of Birmingham, Birmingham, UK ⁴Department of Gastroenterology, Queen Alexandra Hospital Portsmouth, Portsmouth, UK ⁵Department of Gastroenterology, New Cross Hospital, Wolverhampton, UK ⁶Birmingham Clinical Trials Unit, University of Birmingham, Birmingham, UK ⁷Department of Gastroenterology, Basingstoke and North Hampshire Hospital, Basingstoke, UK ⁸West Midlands Regional Genetics Laboratory, Birmingham Women's and Children's NHS Foundation Trust, Birmingham, UK ⁹Department of Gastroenterology, Royal United Hospital Bath, Bath, UK ¹⁰Department of Gastroenterology, Salford Royal Hospital, Manchester, UK ¹¹Department of Gastroenterology, Russells Hall Hospital, Dudley, UK ¹²Department of Gastroenterology, University Hospitals Leicester, Leicester, UK ¹³National Institute for Health Research (NIHR) Birmingham Inflammation Biomedical Research Centre, Birmingham, UK ¹⁴Department of Gastroenterology, Directorate of Medicine and Acute Care, County Durham and Darlington NHS Foundation Trust, Durham, UK ¹⁵Department of Gastroenterology, City Hospital, Birmingham, UK ¹⁶Department of Gastroenterology, King's Mill Hospital, Sutton-in-Ashfield, UK ¹⁷Department of Gastroenterology, Princess Alexandra Hospital, Harlow, UK ¹⁸Department of Gastroenterology, Newport, Isle of Wight, UK ¹⁹Department of Gastroenterology and Inflammatory Bowel Unit, St Mark's Hospital and Academic Institute, Middlesex, UK ²⁰Department of Gastroenterology, Whipps Cross Hospital, London, UK ²¹University Hospitals Birmingham NHS Foundation Trust, Birmingham, UK ²²Department of Gastroenterology, Basildon and Thurrock University Hospital, Essex, UK ²³Department of Gastroenterology, North Tyneside General Hospital, North Shields, UK ²⁴Gastroenterology Unit, John Radcliffe Hospital, Oxford, UK ²⁵Department of Gastroenterology, West Middlesex University Hospital, Middlesex, UK ²⁶Department of Gastroenterology, Royal Bournemouth Hospital, Bournemouth, UK ²⁷Department of Gastroenterology, Blackpool Victoria Hospital, Blackpool, UK ²⁸Department of Gastroenterology, Addenbrookes Hospital, Cambridge, UK ²⁹Department of Gastroenterology, Institute for Translational Medicine, Royal Liverpool and Broadgreen University Hospitals, Liverpool, UK ³⁰Department of Gastroenterology, James Cook University Hospital, Middlesbrough, UK ³¹Department of Gastroenterology, Hull Royal Infirmary, Hull, UK ³²Department of Gastroenterology, Heartlands Hospital, Birmingham, UK ³³Department of Gastroenterology, St James's University Hospital, Leeds, UK ³⁴Department of Gastroenterology, St Mary's Hospital, London, UK ³⁵Department of Gastroenterology, Kettering General Hospital, Kettering, UK *Corresponding author dion.morton@uhb.nhs.uk

Declared competing interests of authors: Ashish Awasthi reports sponsorship to attend the European Crohn's and Colitis Organisation's meeting from Vifor Pharma UK Ltd (Bagshot, UK) and sponsorship to attend British Society of Gastroenterology's meetings from Norgine Ltd (Uxbridge, UK) outside the submitted work. Andrew Beggs reports grants, personal fees and non-financial support from Illumina Inc. (San Diego, CA, USA) and personal fees and other (travel expenses) from Bristol-Myers Squibb (New York, NY, USA) outside the submitted work. Mathew Brookes reports grants and personal fees from Tillotts Pharma (Lincoln, UK), personal fees from AbbVie Inc. (North Chicago, IL, USA) and grants and personal fees from Vifor (International) AG (St Gallen, Switzerland) outside the submitted work. Jon Deeks reports membership of Health Technology Assessment (HTA) Commissioning Boards and Panels (2008 to present) and appointment as a National Institute for Health Research (NIHR) Senior Investigator. Alisa Hart reports personal fees from AbbVie Inc., Dr Falk Pharma (UK) (Bourne End, UK), Ferring (Saint-Prex, Switzerland), Janssen (Beerse, Belgium), Merck Sharp & Dohme Corp. (Kenilworth, NJ, USA), Pfizer (New York, NY, USA), Takeda Pharmaceutical Company Ltd (Tokyo, Japan) and Genetech (San Francisco, CA, USA) outside the submitted work. Laura Magill reports grants from NIHR Efficacy and Mechanism Evaluation during the conduct of the study, and grants from the NIHR HTA and NIHR Research for Patient Benefit programmes outside the submitted work. Dion Morton reports membership of HTA Commissioning Board (2014–17) and appointment as a NIHR Senior Investigator (2015-present). Shaji Sebastian reports a research grant, speaker honoraria, attendance at advisory boards and conference attendance fees from Takeda Pharmaceutical Company Ltd, AbbVie Inc, Merck Sharp & Dohme Corp., Tillotts Pharma, Pharmacosmos (Reading, UK) and Janssen outside the submitted work. Marietta lacucci reports grants and personal fees from Pentax (Tokyo, Japan), grants from Olympus Corporation (Tokyo, Japan) and Fujifilm (Tokyo, Japan) and personal fees from AbbVie Inc. and Janssen outside the submitted work.

Published January 2021 DOI: 10.3310/eme08010

Plain English summary

The ENDCaP-C diagnostic accuracy study

Efficacy and Mechanism Evaluation 2021; Vol. 8: No. 1 DOI: 10.3310/eme08010

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

U lcerative colitis causes damage to the large bowel (lower part of the gut) lining (mucosa) over many years and is associated with an increasing risk of large bowel cancer. Annually, approximately 1000 NHS patients with ulcerative colitis have a major operation to remove the large bowel because of cancerous change. Direct inspection of the bowel (via colonoscopy) is performed regularly to identify early cancer, but this is challenging because the bowel lining appearance is altered by ulcerative colitis. This study looked to develop and test whether or not detecting changes in deoxyribonucleic acid (DNA) (methylation) in the cells lining the large bowel might improve early cancer detection and overcome the difficulty in identifying early cancer.

The first part of the study tested 11 genes (found in the DNA inside every cell) previously reported as being altered in early bowel cancers. This was carried out on small samples of the bowel lining (i.e. biopsies) taken during colonoscopies. Five of the genes tested were identified as being closely associated with early bowel cancer in ulcerative colitis patients.

We then tested the selected five genes in new biopsies from 818 patients (across 31 hospitals) undergoing regular bowel examination for ulcerative colitis. This confirmed an association between these deoxyribonucleic acid changes in background bowel lining and early cancer located elsewhere in the bowel. This suggested that methylation testing might improve the accuracy of the colonoscopy for ulcerative colitis-associated bowel cancer. We further tested this finding by repeating the bowel examination in selected patients. This test did identify more patients with cancerous changes, but adding the methylation test did not show significant benefit above the colonoscopy.

We have shown for the first time that a DNA-based test on the bowel lining can help detect early cancer in patients with ulcerative colitis. However, this association was not strong enough to show a measurable additional benefit.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in man and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

This report

The research reported in this issue of the journal was funded by the EME programme as project number 11/100/29. The contractual start date was in February 2013. The final report began editorial review in September 2018 and was accepted for publication in May 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care. Social Care.

© Queen's Printer and Controller of HMSO 2021. This work was produced by Awasthi *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk